Author's response to reviews

Title: A Fatal Case of Bupropion (Zyban) Hepatotoxicity with Autoimmune Features

Authors:

Fawwaz Humayun (humayuf@hotmail.com)
Thomas M Shehab (shehabt@trinity-health.org)
Joseph A Tworek (itworek@yahoo.com)
Robert J Fontana (rfontana@med.umich.edu)

Version: 4 Date: 1 May 2007

Author's response to reviews:

May 1, 2007

Michael Kidd, MD
Editor-in-Chief
Journal of Medical Case Reports

Dear Dr. Kidd,

We are privileged to be able to submit this case report, "A Fatal Case of Bupropion (Zyban) Hepatotoxicity with Autoimmune Features," to the Journal of Medical Case Reports. This manuscript has not been published nor is it under review elsewhere. We feel this is an extremely important submission given the frequent use of bupropion in a variety of clinical settings and fatal outcome of the patient. In addition, we feel that the clinical presentation will further raise awareness of drug induced liver injury presenting with autoimmune features.

The St. Joseph Mercy Institutional Review Board approved this project.

All authors have fulfilled authorship responsibility.

There are no financial disclosures relevant to this publication. This work was supported in part by a grant to Robert J. Fontana, MD from the National Institutes of Diabetes and Digestive and Kidney Diseases (NIH U01 DK065184-01).

Sincerely,

Thomas M. Shehab, M.D.
Director of Clinical Research
Department of Internal Medicine
St. Joseph Mercy Health System